Author’s response to reviews

Title: Lumbar Puncture as Possible Cause of Sudden Paradoxical Herniation in Patient with Previous Decompressive Craniectomy: Report of Two Cases

Authors:

Liang Shen (dliangshen@163.com)
Sheng Qiu (qius2001@126.com)
Zhongzhou Su (szzmdy@163.com)
Xudong Ma (xudongmahuzhou@162.com)
Renfu Yan (yanrenfu62@163.com)

Version: 3 Date: 12 Jun 2017

Author’s response to reviews:

Comments:

However, there are still some minor language defects such as spelling (for example: "athmospheric") or expression (eg: "in out paper report" in discussion) need to be revised. And also, in your conclusion, I suggest you had better add "CSF drainage during lumbar puncture should be in small volume for patients with DC".

Responses:

Thanks so much for your comments. In our discussion section, the word "athmospheric" is "atmospheric", it seems that there is no spelling errors. Please let us know if we misunderstand your mean. "in out paper report" has been revised as "in our paper report". The sentence that "CSF drainage during lumbar puncture should be in small volume for patients with DC" has been added. Thank you again for your hands.